Title: Cyclin D1, Id1 and EMT in Breast cancer

Authors:

Nicholas P Tobin (tobin.nick@gmail.com)
Andrew H Sims (Andrew.Sims@ed.ac.uk)
Katja L Lundgren (lundgren.katja@gmail.com)
Sophie Lehn (sophie.lehn@med.lu.se)
Goran Landberg (glandberg@picr.man.ac.uk)

Version: 4 Date: 24 September 2011

Author's response to reviews: see over
Manchester, September 24th, 2011

Dear Editor,

Thank you kindly for the opportunity to publish in your journal. We have completed the following formatting changes for our article MS ID: 2115140737536846, as per your requests.

Major formatting changes:

1) Affiliations have been adjusted to reflect requested formatting
2) We have altered our abstract titles in the submission system to reflect our manuscript
3) Additional files have been re-numbered, and in the case of additional files 2 and 4, re-uploaded in the correct order.

Minor formatting changes:

1) All highlighting and tracking has been removed from the manuscript
2) Links to the affiliations have been amended and are now in numerical order
3) All vertical lines have been removed from tables
4) We have completed a spelling and grammar check both manually and by software and have changed any incorrect grammar/spelling.

Kind regards,

Göran Landberg MD, PhD
Breakthrough Breast Cancer Research Unit
School of Cancer and Enabling Sciences
Manchester University
Paterson Institute for Cancer Research
Wilmslow Road
Manchester
M20 4BX, UK
glandberg@picr.man.ac.uk